Author’s response to reviews

Title: A case report of Tinea capitis in infant in first year of life

Authors:

Narcisa Mandras (narcisa.mandras@unito.it)
Janira Roana (janira.roana@unito.it)
Ornella Cervetti (ornella.cervetti@unito.it)
Michele Panzone (mpanzone@cittadellasalute.to.it)
Vivian Tullio (vivian.tullio@unito.it)

Version: 1 Date: 17 Dec 2018

Author’s response to reviews:

Dear Editor,

we have modified the grammatical/format issues in the manuscript as you kindly suggest. However, we have not replaced negativization with sterilization because some opportunist molds were detected on the plates. For greater clarity we changed the sentence in line 76: "the first negativization of the culture for M. canis was observed".

In line 79 we have reported: "After this period, all scalp lesions were completely healed and cultures resulted negative for dermatophytes". Moreover, we added an other reference (no. 12)

As you suggested, the manuscript has been revised by a native English speaker.

Thanking you for the suggestions, we send again a revised version of the manuscript for publication.

Sincerely

Prof. Vivian Tullio